////
Automatically generated by PainlessDocGenerator. Do not edit.
Rebuild by running `gradle generatePainlessApi`.
////

[[painless-api-reference-DateFormat]]++DateFormat++::
** [[painless-api-reference-DateFormat-AM_PM_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#AM_PM_FIELD[AM_PM_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#AM_PM_FIELD[java 9])
** [[painless-api-reference-DateFormat-DATE_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#DATE_FIELD[DATE_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#DATE_FIELD[java 9])
** [[painless-api-reference-DateFormat-DAY_OF_WEEK_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#DAY_OF_WEEK_FIELD[DAY_OF_WEEK_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#DAY_OF_WEEK_FIELD[java 9])
** [[painless-api-reference-DateFormat-DAY_OF_WEEK_IN_MONTH_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#DAY_OF_WEEK_IN_MONTH_FIELD[DAY_OF_WEEK_IN_MONTH_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#DAY_OF_WEEK_IN_MONTH_FIELD[java 9])
** [[painless-api-reference-DateFormat-DAY_OF_YEAR_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#DAY_OF_YEAR_FIELD[DAY_OF_YEAR_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#DAY_OF_YEAR_FIELD[java 9])
** [[painless-api-reference-DateFormat-DEFAULT]]static int link:{java8-javadoc}/java/text/DateFormat.html#DEFAULT[DEFAULT] (link:{java9-javadoc}/java/text/DateFormat.html#DEFAULT[java 9])
** [[painless-api-reference-DateFormat-ERA_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#ERA_FIELD[ERA_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#ERA_FIELD[java 9])
** [[painless-api-reference-DateFormat-FULL]]static int link:{java8-javadoc}/java/text/DateFormat.html#FULL[FULL] (link:{java9-javadoc}/java/text/DateFormat.html#FULL[java 9])
** [[painless-api-reference-DateFormat-HOUR0_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#HOUR0_FIELD[HOUR0_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#HOUR0_FIELD[java 9])
** [[painless-api-reference-DateFormat-HOUR1_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#HOUR1_FIELD[HOUR1_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#HOUR1_FIELD[java 9])
** [[painless-api-reference-DateFormat-HOUR_OF_DAY0_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#HOUR_OF_DAY0_FIELD[HOUR_OF_DAY0_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#HOUR_OF_DAY0_FIELD[java 9])
** [[painless-api-reference-DateFormat-HOUR_OF_DAY1_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#HOUR_OF_DAY1_FIELD[HOUR_OF_DAY1_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#HOUR_OF_DAY1_FIELD[java 9])
** [[painless-api-reference-DateFormat-LONG]]static int link:{java8-javadoc}/java/text/DateFormat.html#LONG[LONG] (link:{java9-javadoc}/java/text/DateFormat.html#LONG[java 9])
** [[painless-api-reference-DateFormat-MEDIUM]]static int link:{java8-javadoc}/java/text/DateFormat.html#MEDIUM[MEDIUM] (link:{java9-javadoc}/java/text/DateFormat.html#MEDIUM[java 9])
** [[painless-api-reference-DateFormat-MILLISECOND_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#MILLISECOND_FIELD[MILLISECOND_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#MILLISECOND_FIELD[java 9])
** [[painless-api-reference-DateFormat-MINUTE_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#MINUTE_FIELD[MINUTE_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#MINUTE_FIELD[java 9])
** [[painless-api-reference-DateFormat-MONTH_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#MONTH_FIELD[MONTH_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#MONTH_FIELD[java 9])
** [[painless-api-reference-DateFormat-SECOND_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#SECOND_FIELD[SECOND_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#SECOND_FIELD[java 9])
** [[painless-api-reference-DateFormat-SHORT]]static int link:{java8-javadoc}/java/text/DateFormat.html#SHORT[SHORT] (link:{java9-javadoc}/java/text/DateFormat.html#SHORT[java 9])
** [[painless-api-reference-DateFormat-TIMEZONE_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#TIMEZONE_FIELD[TIMEZONE_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#TIMEZONE_FIELD[java 9])
** [[painless-api-reference-DateFormat-WEEK_OF_MONTH_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#WEEK_OF_MONTH_FIELD[WEEK_OF_MONTH_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#WEEK_OF_MONTH_FIELD[java 9])
** [[painless-api-reference-DateFormat-WEEK_OF_YEAR_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#WEEK_OF_YEAR_FIELD[WEEK_OF_YEAR_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#WEEK_OF_YEAR_FIELD[java 9])
** [[painless-api-reference-DateFormat-YEAR_FIELD]]static int link:{java8-javadoc}/java/text/DateFormat.html#YEAR_FIELD[YEAR_FIELD] (link:{java9-javadoc}/java/text/DateFormat.html#YEAR_FIELD[java 9])
* ++[[painless-api-reference-DateFormat-getAvailableLocales-0]]static <<painless-api-reference-Locale,Locale>>[] link:{java8-javadoc}/java/text/DateFormat.html#getAvailableLocales%2D%2D[getAvailableLocales]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getAvailableLocales%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-getDateInstance-0]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getDateInstance%2D%2D[getDateInstance]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getDateInstance%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-getDateInstance-1]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getDateInstance%2Dint%2D[getDateInstance](int)++ (link:{java9-javadoc}/java/text/DateFormat.html#getDateInstance%2Dint%2D[java 9])
* ++[[painless-api-reference-DateFormat-getDateInstance-2]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getDateInstance%2Dint%2Djava.util.Locale%2D[getDateInstance](int, <<painless-api-reference-Locale,Locale>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#getDateInstance%2Dint%2Djava.util.Locale%2D[java 9])
* ++[[painless-api-reference-DateFormat-getDateTimeInstance-0]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getDateTimeInstance%2D%2D[getDateTimeInstance]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getDateTimeInstance%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-getDateTimeInstance-2]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getDateTimeInstance%2Dint%2Dint%2D[getDateTimeInstance](int, int)++ (link:{java9-javadoc}/java/text/DateFormat.html#getDateTimeInstance%2Dint%2Dint%2D[java 9])
* ++[[painless-api-reference-DateFormat-getDateTimeInstance-3]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getDateTimeInstance%2Dint%2Dint%2Djava.util.Locale%2D[getDateTimeInstance](int, int, <<painless-api-reference-Locale,Locale>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#getDateTimeInstance%2Dint%2Dint%2Djava.util.Locale%2D[java 9])
* ++[[painless-api-reference-DateFormat-getInstance-0]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getInstance%2D%2D[getInstance]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getInstance%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-getTimeInstance-0]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getTimeInstance%2D%2D[getTimeInstance]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getTimeInstance%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-getTimeInstance-1]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getTimeInstance%2Dint%2D[getTimeInstance](int)++ (link:{java9-javadoc}/java/text/DateFormat.html#getTimeInstance%2Dint%2D[java 9])
* ++[[painless-api-reference-DateFormat-getTimeInstance-2]]static <<painless-api-reference-DateFormat,DateFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getTimeInstance%2Dint%2Djava.util.Locale%2D[getTimeInstance](int, <<painless-api-reference-Locale,Locale>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#getTimeInstance%2Dint%2Djava.util.Locale%2D[java 9])
* ++[[painless-api-reference-DateFormat-getCalendar-0]]<<painless-api-reference-Calendar,Calendar>> link:{java8-javadoc}/java/text/DateFormat.html#getCalendar%2D%2D[getCalendar]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getCalendar%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-getNumberFormat-0]]<<painless-api-reference-NumberFormat,NumberFormat>> link:{java8-javadoc}/java/text/DateFormat.html#getNumberFormat%2D%2D[getNumberFormat]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getNumberFormat%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-getTimeZone-0]]<<painless-api-reference-TimeZone,TimeZone>> link:{java8-javadoc}/java/text/DateFormat.html#getTimeZone%2D%2D[getTimeZone]()++ (link:{java9-javadoc}/java/text/DateFormat.html#getTimeZone%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-isLenient-0]]boolean link:{java8-javadoc}/java/text/DateFormat.html#isLenient%2D%2D[isLenient]()++ (link:{java9-javadoc}/java/text/DateFormat.html#isLenient%2D%2D[java 9])
* ++[[painless-api-reference-DateFormat-parse-1]]<<painless-api-reference-Date,Date>> link:{java8-javadoc}/java/text/DateFormat.html#parse%2Djava.lang.String%2D[parse](<<painless-api-reference-String,String>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#parse%2Djava.lang.String%2D[java 9])
* ++[[painless-api-reference-DateFormat-parse-2]]<<painless-api-reference-Date,Date>> link:{java8-javadoc}/java/text/DateFormat.html#parse%2Djava.lang.String%2Djava.text.ParsePosition%2D[parse](<<painless-api-reference-String,String>>, <<painless-api-reference-ParsePosition,ParsePosition>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#parse%2Djava.lang.String%2Djava.text.ParsePosition%2D[java 9])
* ++[[painless-api-reference-DateFormat-setCalendar-1]]void link:{java8-javadoc}/java/text/DateFormat.html#setCalendar%2Djava.util.Calendar%2D[setCalendar](<<painless-api-reference-Calendar,Calendar>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#setCalendar%2Djava.util.Calendar%2D[java 9])
* ++[[painless-api-reference-DateFormat-setLenient-1]]void link:{java8-javadoc}/java/text/DateFormat.html#setLenient%2Dboolean%2D[setLenient](boolean)++ (link:{java9-javadoc}/java/text/DateFormat.html#setLenient%2Dboolean%2D[java 9])
* ++[[painless-api-reference-DateFormat-setNumberFormat-1]]void link:{java8-javadoc}/java/text/DateFormat.html#setNumberFormat%2Djava.text.NumberFormat%2D[setNumberFormat](<<painless-api-reference-NumberFormat,NumberFormat>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#setNumberFormat%2Djava.text.NumberFormat%2D[java 9])
* ++[[painless-api-reference-DateFormat-setTimeZone-1]]void link:{java8-javadoc}/java/text/DateFormat.html#setTimeZone%2Djava.util.TimeZone%2D[setTimeZone](<<painless-api-reference-TimeZone,TimeZone>>)++ (link:{java9-javadoc}/java/text/DateFormat.html#setTimeZone%2Djava.util.TimeZone%2D[java 9])
* Inherits methods from ++<<painless-api-reference-Format,Format>>++, ++<<painless-api-reference-Object,Object>>++
